# Review of Human–Machine Interfaces for Small Unmanned Systems With Robotic Manipulators

Sierra N. Young<sup>(D)</sup>, *Member*, *IEEE*, and Joshua M. Peschel<sup>(D)</sup>, *Member*, *IEEE* 

Abstract—This article reviews the human-machine interaction (HMI) technologies used for telemanipulation by small unmanned systems (SUS) with remote manipulators. SUS, including land, air, and sea vehicles, can perform a wide range of reconnaissance and manipulation tasks with varying levels of autonomy. SUS operations involving physical interactions with the environment require some level of operator involvement, ranging from direct control to goal-oriented supervision. Telemanipulation remains a challenging task for all levels of human interaction because the operator and the vehicle are not colocated, and operators require HMI technologies that facilitate manipulation from a remote location. This article surveys the human operator interfacing for over 70 teleoperated systems, summarizes the effects of physical and visual interface factors on user performance, and discusses these findings in the context of telemanipulating SUS. This article is of importance to SUS researchers and practitioners who will directly benefit from HMI implementations that improve telemanipulation performance.

*Index Terms*—Human-robot interaction, manipulators, telerobotics, unmanned aerial vehicles, user interfaces.

### I. INTRODUCTION

T HIS article surveys the human-machine interaction (HMI) technologies for small unmanned systems (SUS) equipped with robotic manipulators capable of performing telemanipulation tasks. SUS have historically been used for visual reconnaissance, but needs beyond observation expanded research efforts toward physical interaction with remote environments [1]. Telemanipulation capabilities of SUS enable a greater understanding of the remote environment through physical contact, and are particularly advantageous when tasks are too dangerous, complex, costly, or difficult for humans to perform [2]. Domain-specific examples of SUS telemanipulation include construction material transport with unmanned aerial vehicles (UAVs) [3]–[5], space exploration and sampling with unmanned ground vehicles (UGVs) [6]–[8], underwater sampling with remotely operated vehicles (ROVs) [9], [10], and aquatic sampling with unmanned

Manuscript received April 17, 2019; revised August 15, 2019 and October 22, 2019; accepted December 12, 2019. Date of publication February 24, 2020; date of current version March 12, 2020. The work of S. N. Young was supported in part by the United States Department of Defense, in part by the Air Force Office of Scientific Research, and in part by the National Defense Science and Engineering Graduate under Fellowship 32 CFR 168a. This article was recommended by Associate Editor J. A. Adams. (*Corresponding author: Joshua M. Peschel.*)

S. N. Young is with the Department of Biological and Agricultural Engineering, North Carolina State University, Raleigh, NC 27695-7625 USA (e-mail: syoung2@ncsu.edu).

J. M. Peschel is with the Department of Agricultural and Biosystems Engineering, Iowa State University, Ames, IA 50011-3270 USA (e-mail: peschel@iastate.edu).

Digital Object Identifier 10.1109/THMS.2020.2969380

surface vehicles (USVs) [11]–[13]. SUS can also perform specialized tasks such as bomb disposal [14] and maintenance and service tasks [15]. These examples highlight the applications of SUS capable of remote manipulation, which range from conducting unique scientific explorations to performing repetitive practical tasks.

Research trends in perception [16], motion planning [17], and real-time control [18] indicate that the autonomous capabilities of SUS are improving; however, performing fully autonomous manipulation tasks in real-world environments still remains a challenge [19], [20]. A telemanipulation task in the context of SUS operations is any action that requires physical interaction with the remote environment with a robotic manipulator, based on control input that arrives through teleoperation. Teleoperation is the act of using a machine that extends input from a human operator to an SUS in a remote location, and the operator control input can range from direct human control to supervisory control. Performing telemanipulation tasks from a teleoperated SUS will require some level of human input; therefore, it is important to understand how HMI technology affects telemanipulation performance.

An SUS includes the vehicle (or robot), which may be a ground, aerial, surface, or tethered underwater vehicle, a ground control station, an operator, and a communication link. "Small" unmanned systems are systems with vehicles that are selfcontained and portable by humans. The vehicle in the context of telemanipulation must include a robotic manipulator capable of performing a manipulation task. This article considers the HMI for SUS that perform general telemanipulation tasks in relatively small workspaces and in shorter time periods (i.e., minutes, not hours). This article focuses only on the telemanipulation task and does not include navigation tasks. The types of remote tasks considered in this article include pick-and-place, grasping, probing, pulling, and pushing tasks. This article excludes teleoperated systems with unusually large workspaces (e.g., excavation vehicles), prohibitively long time delays (e.g., space operation), or specialized training requirements (e.g., telesurgery).

Teleoperated SUS with robotic manipulators operate on a spectrum of autonomous capabilities. This article includes system autonomy in the analysis of human–machine interfacing because the robot's level of autonomy affects the overall human-robot interaction [21], [22]. There are multiple autonomy taxonomies that appear in the literature [22]–[24]; however, this article considers autonomy to be the extent to which a robot can sense the environment, plan based on that environment, and act upon that environment, with the intent of reaching

This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see http://creativecommons.org/licenses/by/4.0/

some goal without external control [22]. This article considers the following four levels of autonomy [25], [26]: remote control, human-assisted, human-delegated, and human-supervised. These categories are described in greater detail below (note that fully autonomous systems are not considered in this analysis as human operators are not involved with the telemanipulation task and can be ignored by the system [23]).

## A. Remote Control Systems

Remote control occurs when the operator interface represents all manipulator degrees of freedom, and the system operator directly controls the manipulator actuators [25], [26]. Remote control is either noncomputer-aided, where operator inputs map directly to manipulator outputs, or computer-aided, where computer transformation occurs prior to sensing or actuating the manipulator [26]. Both cases are remote control operation because there is no system autonomy, and the system only performs manipulation tasks when there are control inputs. An example of remote control telemanipulation is using a hardware interface to directly control a manipulator attached to a tethered ROV to perform geological sampling in deep sea environments.

## B. Human-Assisted Systems

Human-assisted teleoperation is a form of shared control [24], [26]. Under human-assisted telemanipulation, the operator manually controls a larger proportion of manipulator functions than the system can autonomously perform; therefore, the human is the ultimate decision-maker [25]. An example of human-assisted telemanipulation is when a UGV manipulator operator selects a desired object within an interface, confirms a sequence of actions, and the robot automatically grasps the object.

#### C. Human-Delegated Systems

Human-delegated teleoperation is also a form of shared control [24], [26]. In human-delegated systems, the SUS maintains more control than the operator, and the operator provides guiding commands that use autonomous functionality within the system [25]. An example of human-delegated telemanipulation is the confirmation of objects to retrieve after autonomous detection by the SUS, or the confirmation of manipulator trajectories autonomously computed by the SUS and provided to the operator.

## D. Human-Supervised Systems

In a human-supervised system, the operator provides highlevel goal-oriented commands, and the decision-making capabilities and manipulator control lie with the SUS [25]. Human-supervised systems receive goal-oriented commands, autonomously gather and process sensory data, make decisions, and carry out actions, all while informing the operator of its status. For example, a human-supervised UGV may continuously perform a fully autonomous grasping task and will only alert the human operator when it encounters a problem. This article applied the following inclusion criteria to collect appropriate references.

- The system was an unmanned ground, aerial, surface, or tethered underwater vehicle equipped with a robotic manipulator, or a teleoperated manipulator similar in spatial and temporal scales and task compared to SUS telemanipulation.
- 2) This article focused on the telemanipulation task.
- This article provided empirical or experimental evidence related to operator performance when using a specified HMI technology for telemanipulation.
- The task type included some form of physical interaction between the manipulator and the remote environment.

Applying these inclusion criteria resulted in references from the UAV, UGV, ROV, and USV literature, as well as literature from the broader human–robot interaction community when the scope of the telemanipulation task was similar to SUS telemanipulation. Data and results presented in this article include the observations and task performance metrics reported in the included references.

The rest of this article is organized as follows. Section II describes common telemanipulation operator interface technologies. Section III describes physical factors that affect telemanipulation, and Section IV describes types of visualization that affect telemanipulation performance. Finally, Section V concludes this article.

# II. OPERATOR INTERFACES USED FOR TELEMANIPULATION INPUTS

This section describes interface types used for operator telemanipulation inputs, including semiautomatic controls, masterslave devices, computer workstations, and mobile tablets.

#### A. Master-Slave Controllers

A typical telemanipulation input scheme is a master–slave configuration where the master controller is geometrically similar to the manipulator at certain scales [27]. In this setup, operator inputs from the master arm map directly to the slave arm under remote control operation, and the position and force responses of both master and slave devices may be identical or scaled proportionally, depending on the task and system. The master controller can control each degree offreedom individually, or a subset of the degrees of freedom. Additionally, the master is often a kinematic replica of the slave, providing an intuitive interface [24].

For SUS operation, tethered ROVs normally use master–slave controllers to perform subsurface manipulation. Commercially available manipulators, such as the seven-degrees-of-freedom Orion (Schilling Robotics, USA), are commonly deployed with ROVs due to their robustness in deep-sea environments and readily available master controllers [9], [28]–[30]. The master controller for these systems is a scaled-down kinematic replica attached to a console panel with push keys, a small screen, and status indicators. Master joint movements produce an equivalent or scaled movement in the slave arm, and the keys determine manipulator control mode (e.g., rate or position) and functions (e.g., freezing or enabling hydraulics).

#### **B.** Semiautomatic Input Devices

Semiautomatic input devices enable remote control and human-assisted control during telemanipulation tasks. These devices have geometries dissimilar to the manipulator [27], such as joysticks or other physical controllers. Joysticks can operate under either position or rate control; under position control the joystick commands the desired position of the joint or end effector, and under rate control the joystick commands the desired velocity. In both cases, the input commands are proportional to the joystick displacement [24]. Dual joysticks (e.g., gamepad controllers) provide input if the slave manipulator requires control for both orientation and translation [24]. Joysticks and gamepad controllers have operated manipulators on ROVs [31], [32], UGVs [33], and simulated UAVs [34].

Nonjoystick, three-dimensional (3-D) semiautomatic devices can also be used for telemanipulation input under direct and human-assisted control. Examples include the Novint Falcon haptic device (Novint Technologies LLC, USA), a threedegrees-of-freedom version of the original delta-robot configuration [35], and the Geomagic Touch<sup>TM</sup> (3D Systems, USA). These smaller, inexpensive master controllers are widely studied for applications in robotics and gaming [36], [37] and can provide force feedback based on user input. These controllers normally have smaller working volumes compared to traditional master–slave configurations.

#### C. Computer Workstations

Operators can control teleoperated systems via communication networks through computer workstations, including laptops and desktops. Networked SUS can be controlled through user interface software implemented on a workstation, which requires a keyboard and/or mouse for user input [38]. For remote control or human-assisted operation, operators can control the manipulator by deciding where to click within the interface or through keystroke inputs, which control manipulators by position and may require mapping 2-D user input to locations in 3-D space. Full 3-D representation in remote control requires interface elements that control each degree of freedom separately (e.g., control in the x-, y-, and z-axes [39]) or control the position of the end effector through inverse kinematics [40]. In general, workstations can process and display large amounts of data and complex mixed visualizations [41]–[43]; however, one constraint of workstations is providing an adequate 2-D representation of the manipulator and robot that exist in a 3-D remote environment. Adequate 2-D representations are normally attempted through visualizations such as video streams [44], [45], stereo vision [46], [47], or mixed reality (MR) displays [6], [48]. Additionally, high-level directives can be issued from a computer workstation to facilitate human-delegated or humansupervised operation.

## D. Mobile Tablets

Mobile tablets are an appealing interface technology to use for telemanipulation control because they are widely available, are low cost, and use standard gestures [49]–[51]. Operators can give either high level directives or control the manipulator position directly through tablet interfaces, enabling support for a wide range of autonomy levels. Touch-based interfaces are generally intuitive for nonexpert users [51], but the interface elements that control the manipulation affect operator performance under remote control operation. The development of telemanipulation interfaces for these handheld devices is becoming increasingly important; however, challenges remain, including intuitively mapping 2-D inputs to 3-D space [44], decreased computational power and the need for efficient algorithms [42], [52], and designing effective gesture-based touch controls [51]. Additionally, mobile device visualizations are less complex compared to large workstations, as bandwidth constraints limit how much sensor data and information can stream to and be processed on the device.

#### **III. EFFECTS OF INPUT DEVICE ON TELEMANIPULATION**

This section discusses the types of physical factors that affect SUS operator performance during telemanipulation, including controller form factors and feedback variations of manual control. Table I presents an overview of the physical interface implementations surveyed in this section. The results from telemanipulation HMI that were evident across multiple studies included in this section were synthesized into common themes, or findings, that are presented in Table II.

#### A. Input Form Factor

Control input devices are a critical link between operators and teleoperated systems in remote environments [53], [54]. The form factor and design of input devices influence intuitiveness and learning requirements, which subsequently affect operator performance [33], [34]. The sections below discuss how telemanipulation interfaces affect operator performance, and includes the following interface devices—physical hand controllers, workstations, and touch-based mobile devices.

1) Physical Hand Controllers: Common telemanipulation hand controllers include master–slave devices, joysticks, and other 3-D devices. Depending on the level and location of control within the manipulator, each design affords different types of interaction between the controller device and robot. For example, hand controllers that are geometrically similar and retain all robotic manipulator degrees of freedom are normally easier for users to learn because they intuitively map user input to output [9], [33]. A remote operator identifies their body and immediate environment with the remote vehicle and its environment; therefore, a geometrically similar device better enables the match between their own movements and the remote manipulator attached to the vehicle [55].

Joysticks are also commonly used input devices, but because they are geometrically dissimilar to manipulators, visual motor mapping between the end effector and joystick affects task completion [56], [57]. Situation awareness of the vehicle's orientation in the surrounding environment is necessary to achieve proper mental mapping between the SUS manipulator and joystick [58]. To improve performance, joystick input should be analogous (i.e., moving the joystick to the left

TABLE I SUMMARY OF PHYSICAL INTERFACE IMPLEMENTATIONS FOR TELEMANIPULATION

	Input Device							Control				Level of Autonomy			
First Author (Ref.)	Domain	Master-Slave	Joystick	Mouse	Keyboard	Buttons	Touch Device	End-Effector	Joint Position	Bilateral Feedback	Virtual Fixtures	Remote Control	Human-Assisted	Human-Delegated	Human-Supervised
Atherton (2009) [63]	RM		٠					•	٠			•			
Bengel (2009) [64]	UGV						•							•	
Chen <sup>1</sup> (2013) [41]	UGV			•	٠			•					•	٠	
Hashimoto (2011) [50]	UGV						٠		•			•			
Herbert (2015) [61]	UGV						•							•	
Herdocia (2012) [65]	RM	•							•			•			
Kuang (2004) [66]	RM	•			•			•			•		•		
Lee (2006) [67]	ROV		•		•			•				•			
Lopez (2014) [51]	UGV						٠		•			•			
Lunghi (2018) [62]	UGV		٠	•	٠			•	•			•			
Materna <sup>1</sup> (2017) [43]	UGV			•	•		•	•					•	•	
Navarro (2015) [68]	RM	•							•	•		•			
Nawab (2007) [48]	RM		•					•				•			
Nixon (2015) [34]	UAV	•	٠						•			•			
Parga (2013) [52]	RM						٠	•				•			
Payandeh (2002) [69]	RM	•						•			٠		•		
Saito (2002) [70]	UGV			•	٠		٠							٠	
Shim <sup>1</sup> (2010) [9]	ROV		٠	•	٠			•	•			•	٠	٠	
Singh (2013) [60]	RM					٠	٠	•				•			
Vozar (2012) [33], [71]	UGV	•	•						•			•			
Xia <sup>1</sup> (2012) [72]	RM	•				•			•		•	•	•		
Yip (2011) [73]	RM	•							•	•		•			
You <sup>1</sup> (2011) [59]	RM			•	•			•	٠			•	•		

 $^{1}$ Some systems had multiple possible control modes that were included in this table.  $^{2}$ RM = Remote manipulator. These systems were teleoperated manipulators similar in spatial and temporal scales to SUS telemanipulation using compatible interface types.

TABLE II

SUMMARY OF THE EFFECTS OF MULTIPLE PHYSICAL FACTORS ON TELEMANIPULATION PERFORMANCE

Physical Factor	Summary of Findings
Type of Hardware	
Physical Hand Controllers	Geometrically similar master controllers can be easier to use, are more intuitive, and improve performance compared to joysticks; if using joysticks, inputs and outputs should be analogous. [9], [33], [34], [44], [54], [56]
Workstation Devices	Keyboard or button-only interfaces can negatively impact performance, but 'click-and-drag' and 'point-and-click' inputs are effective with computer mice. [41], [43], [59], [60]
Touch Devices	Precise control is possible, but can be difficult to achieve; touch devices are effective for programmatically performing sequences of behaviors to complete tasks. [50], [51], [61], [64], [70]
Direct Control Feedback Type	
Position Control, No Feedback	Joysticks offer better performance when operated under position instead of rate control; should use end- effector position control when emphasizing precision; operator fatigue is a potential issue when controlling large master controllers. [53], [63], [74]–[78]
Bilateral Feedback	Can decrease task completion time, reduce error rate, and magnitude of applied forces; does not necessarily improve performance under control time delays. [68], [73], [79]–[81]
Virtual Fixtures	Can increase precision, reduce mental workload, and decrease task completion time; must know remote environment <i>a priori</i> or model it correctly in real-time to be effective. [66], [69], [72], [82]–[84]

moves the vehicle to the left); however, this becomes difficult as manipulators increase in complexity and planer movements do not correspond directly to responses in the system. Joysticks may not be as intuitive as master–slave devices for direct joint control, but they can be effective for end-effector control [9].

Some studies have specifically compared the effects of using joysticks versus master controllers as input devices for telemanipulation tasks. Nixon *et al.* [34] found that joystick input increased task completion time compared to a 3-D input device when performing a simulated UAV manipulation task. Vozar *et al.* [33], [44] studied the effects of a master–slave device compared to dual joysticks for teleoperating a UGV manipulator and found that a master–slave device was more effective than a traditional gamepad controller [33]. These results are consistent with previous findings that position master–slave configurations are more suitable for dexterous manipulation because they have a natural correspondence in time and space as the operator performs movements [53], [55].

2) Workstation Input Devices: Workstation input devices, such as keyboard and mice, facilitate control of systems that have higher levels of automation and complexity. Keyboards and mice are also widely available physical interfaces that nonexpert operators are likely accustomed to using. One limitation, however, is the immobility of networked workstations that require operation in close proximity to their operator.

Common input modes for workstation control include "clickand-drag," where the user clicks a point on the robot on a screen and drags the cursor to indicate desired translation, or "point-and-click," where the user defines a goal end-effector position by clicking within the remote environment [41]. You and Hauser [59] compared multiple click-and-drag mouse input schemes for remote manipulator control, including direct joint control, inverse kinematics, reactive potential field, and a real-time motion planner. The real-time motion planning strategy reduced task completion time and was rated favorably by users because it allowed users to focus more on manipulator positioning, instead of on constructing paths.

A potential issue when using a workstation and 2-D mouse for manipulator control input is precise positing in a 3-D remote environment. Technologies, such as 3-D mice, can be used to assist with precise cursor positioning. Materna *et al.* [43] developed an interface that used a 2-D mouse to interact with and set the 3-D scene, and a 3-D mouse set the end-effector goal pose within that space. Users found the 3-D mouse to be more comfortable and intuitive, and it enabled more precise manipulation of the end effector.

Buttons, including keyboard strokes or clicking buttons with a mouse on a workstation interface, should primarily generate sequences of actions (e.g., behavior chaining), rather than directly control manipulators [45], [60], [61]. The operator control unit for the SURROGATE ground vehicle ran on a workstation desktop and was operated with a standard keyboard and mouse. Using this interface, operators sent entire sequences of actions by chaining behaviors together to reduce operator interaction time [61]. Using keyboards and buttons for direct control can degrade task performance, although some operators prefer the kinesthetic feedback from physically pressing buttons [60], which can result in more accurate performance when precise movements are required [62].

3) Touch Input Devices: A variety of 2-D touch elements enable the operation of manipulators in 3-D environments. Lopez *et al.* [51] created a touch-based version of direct control hardware. Their interface contained four control element options, including virtual buttons, virtual joysticks, touch-screen gestures, and tilt gestures. Touch-based buttons and joysticks yielded the best task completion scores because they enabled more precise control of manipulator movements (compared to the tilt and touch gesture interfaces).

Other tablet interface implementations have taken a direct manipulation design approach, where users can touch the screen to control the robot "directly" in the remote environment (similar to a click-and-drag input scheme). Hashimoto *et al.* [50] developed a touch-screen application in which users directly controlled a UGV manipulator by touching it on a view of the world, as seen from a third person view obtained by pointing a camera at the robot. While the touch controls were intuitive, some users requested a stylus pen to enable more precise interaction, and others reported it was difficult to understand the depth of the space from a single view. Singh *et al.* [60] also developed a direct touch-screen control method for a remote manipulator where operators could touch and drag the manipulator end effector in a desired trajectory, which reduced task completion time and mental workload.

While touch-based gestures may be effective, limited bandwidth and computational power limit the use of advanced data visualizations and control algorithms on tablets [42]. Mobile tablets only afford simplified data visualizations and interaction elements as it is impractical to send and display all robot and sensor data [42]. Additionally, when designing kinematic algorithms for touch-and-drag control (which is often preferred by users [50], [60]), they must be computationally efficient in order to run on tablet devices. For example, Parga *et al.* [52] and Singh *et al.* [60] developed touch-based interfaces for a remote manipulator with inverse kinematic solvers capable of running on small tablet devices.

Direct control can be difficult on a touch-based 2-D device due to a lack of input precision [50], [51], but operators can more readily program behavior and action sequences when using a mobile device. Bengel et al. [64] developed a pictorial tablet interface that enabled human-delegated inspection tasks by "teaching" the UGV desired navigation and inspection tasks, after which the robot autonomously completed the same mission. Herbert et al. [61] developed a tablet interface for controlling a UGV equipped with two manipulators. Their interface only needed high-level directives and relied on autonomy to carry out individual actions by first selecting the object in the image, followed by selecting the appropriate behavior sequence. Saito and Suehiro [70] developed the TeachIng Tablet Interface with a 2-D input screen to enable pick-and-place operations by a portable manipulator. Operators drew pick-up and place frames on the 2-D screen and assigned them to predefined trajectories for the system to complete, effectively teaching the robot to perform manipulation.

#### B. Feedback Variations of Manual Control

Methods for mapping control inputs to outputs affect performance when the operator has some level of manual control of the manipulator [33], [53], [63] (e.g., remote control or human-assisted). This section includes the following variations of feedback for manual control—direct control with no feedback (including both position and rate control), bilateral control (haptic feedback), and virtual fixtures (a type of force feedback). Additionally, this section includes information on the location of control in the manipulator, for example, if the user controls only the end effector or has command over individual degrees of freedom.

1) Direct Control With No Feedback: Under direct position control, operators control each joint or degree-of-freedom of the manipulator individually, and a reference position maps from the input device to the output device [53]. Master-slave and joystick interfaces are commonly used for direct control and can operate by controlling either the position or rate of each joint. Early studies analyzed the effects of position and rate control on target acquisition and found that using position controlled joysticks resulted in improved performance compared to velocity controlled joysticks [74]–[76]. Experiments by Kim et al. [77], [85] examined pick-and-place tasks using two isotonic joysticks to control a remote manipulator and also found better operator performance under position control. If the control device is slow, however, superior performance of position control diminishes, and it is recommended to use position control for small workspace tasks and rate control for slow-workspace tasks [77].

Whether the operator has control over the end effector or each joint also affects operator performance, and the optimal location of control depends on the importance of speed versus accuracy on the task. Atherton and Goodrich [63] found that participants worked faster to control a remote manipulator with a joystick under joint control compared to end-effector control, but joint control resulted in more collisions with the environment. Additionally, operators may be imprecise in positioning the end effector at the desired location and tended to overshoot the target when operating a master-slave controller under position control [78]. Draper [53] also notes that when the operator is in constant manual control of a remote manipulator, the user should be in control of the end effector as direct joint control can be inefficient. For example, inverse kinematics can compute manipulator joint angles to give the operator direct end-effector control [9], [67].

Achieving transparency under position control has traditionally been the goal of master–slave manipulator systems [86]. Master–slave kinematic laws have generalized this type of system for mixed position and rate control [87]; however, when controlling both manipulator position and rate, the process for switching between these modes affects user performance [54], [65]. Herdocia *et al.* [65] found that a manual switch between position and rate control when operating a remote manipulator resulted in slower performance and higher error rates than a differential-end-zone coordination scheme. In their differentialend-zone coordination scheme, a fixed inner space controlled position, and velocity control occurred at the boundary of the inner space.

Operator fatigue is another issue that can arise when operating master-slave devices [53], [78]. Operator fatigue under remote control can degrade operator performance due to controller complexity [88]. For example, Gupta et al. [78] found that operators became tired after repeating thirty pick-and-place tasks on average when using a master controller, despite that a majority of users became proficient within a few minutes. Position control enhancements can reduce fatigue in master-slave operators. Love and Book [89] found that an adaptive impedance manipulator controller reduced the total energy output of the operator when executing remote manipulation tasks compared to nonadaptation. Force scaling can also reduce the required magnitude of operator input by scaling input forces up when operating large devices, or scaling down when performing micromanipulations [86]. Implementing smaller working volumes for the master controller also reduces operator fatigue [90], and fatigue likely decreases after being adequately trained with the master controller [91].

2) Bilateral Force Feedback: Force feedback (also known as haptic feedback) is a broad term that includes both tactile and kinesthetic information. In general, forces felt by the robot using tactile and other sensors are then fed to a haptics device to provide force feedback to a user [92]. Feedback and additional cues from force and tactile sensors in teleoperation systems can complement visual information, increase spatial awareness of the remote environment, and reduce error rate and magnitudes of applied forces [80], [93].

The presence of haptic feedback can decrease completion time. Implementing haptic feedback with proximity sensors for remote grasping tasks reduced completion time compared to visual-only feedback, despite that users were not requested to be time-efficient [68]. Haptic feedback can also reduce average and maximum applied forces [73], [81]. Haptic feedback in a master controller reduced contact forces and the occurrence of large robot-environment interaction forces during telemanipulation [73]. Acceleration haptic feedback can significantly reduce peak and average contact forces when grasping flexible objects [81]. A reduction in manipulator force is especially important for operations when the manipulation target is highly sensitive to force. In general, haptic feedback with small workspace controllers can often offer performance benefits for direct or human-assisted control operations for short-range SUS deployments when communication delays between the operator and vehicle are not significant.

3) Virtual Fixtures: Virtual fixtures are task-dependent computer-generated guides overlaid on a remote workspace reflection [94]. Virtual fixtures provide force feedback similar to general haptics discussed in the previous section; however, virtual fixtures assist the operator with force feedback for a specific, structured teleoperation task. Examples of virtual fixture implementations include a guide from the robot gripper to an object in the remote workspace [69] or force clues influencing the trajectory of the gripper [66].

Virtual fixtures are either "guidance" or "forbidden region": *guidance* virtual fixtures assist in keeping the manipulator on desired paths, while *forbidden region* virtual fixtures physically prevent motion in the remote workspace in specific forbidden

		Visualization Representation				Data and Sensors			Device		Level of Autonomy			
First Author (Year)	Domain	Real Environment	Augmented Reality	Augmented Virtuality	Virtual Environment	Video Feed (RGB)	Stereoscopic Imagery	Laser/Range Scanning	Touchpad Device	Workstation	Remote Control	Human-Assisted	Human-Delegated	Human-Supervised
Afaq (2015) [113]	UGV	•				•				٠	٠			
Atherton (2009) [63]	$\mathrm{R}\mathrm{M}^1$			•		•		•		•	•			
Chintamani (2010) [103]	RM		•			•				•	•			l
Day (2008) [14]	UGV	•					•			•	•			
Hashimoto (2011) [50]	UGV		٠			•			•		•			
Hine (1994) [10]	ROV				•		•			•	•	•	•	
Marani (2009) [114]	ROV				•	•		•		•	•			٠
Moore (2016) [46]	UGV			٠		•	•	•		•	٠		•	
Nawab (2007) [48]	RM		•			•				•	•			
Nguyen (2001) [6]	UGV			٠			•			•	•		•	
Shim (2010) [9]	ROV				٠					•		•	•	
Thomas (2011) [101]	UGV		•			•	•			•		•		
Vozar (2012) [33]	UGV		٠		•	•				•	•			
Wilcox (2007) [115]	UGV	•				•	•			•	•	•		
Zhang (2003) [116]	ROV				•	•	•			•	•		•	

 TABLE III

 SUMMARY OF VISUAL INTERFACE IMPLEMENTATIONS FOR TELEMANIPULATION

<sup>1</sup>RM = Remote Manipulator. These systems were teleoperated manipulators similar in spatial and temporal scales to SUS telemanipulation using compatible interface types.

zones [84] (note that "guiding" fixtures are likely better at micro scales due to slight inaccuracies in system control [95]). Virtual fixtures have been successful in guiding remote control operations, providing localized references, reducing mental workload, and increasing precision [94], [96].

Studies show that virtual fixtures can reduce task completion time for manipulation and positioning tasks [69], [72]. Payandeh and Stanisic [69] reduced task completion time for a remote manipulator through a human-assisted control scheme where the operator positioned the gripper close to a position fixture, and the system autonomously completed the task using the virtual force cues. Kuang *et al.* [66] similarly found improved positioning of a remote manipulator with the assistance of haptic and graphic virtual fixtures.

One limitation of implementing virtual fixtures, however, is their dependence on *a priori* knowledge of the nature and geometry of the task. The implementation of virtual fixtures works well for controlled tasks, such as path following or manufacturing; however, SUS deployment environments are often dynamic and uncertain, and obtaining *a priori* knowledge of the remote environment is difficult. One potential solution is to use remote sensing methods to develop graphical models of the remote environment at the site of manipulation in real time [72], [97], which is only possible if the system is accurately modeled. In changing environments, adaptive virtual fixtures can help overcome new obstacles [98]; however, these systems require training data sets to learn desired paths prior to dynamic environmental changes. More research is needed regarding the modeling and development of virtual fixtures in real time in

unstructured environments before the benefits of virtual fixtures can be fully realized for SUS operations.

## IV. TYPES OF VISUAL DISPLAYS THAT AFFECT TELEMANIPULATION

This section discusses multiple types of visual displays that affect SUS operator performance during telemanipulation. Common remote manipulation displays include MR [including augmented reality (AR) and augmented virtuality (AV)], virtual reality, and stereo vision/depth imagery. Table III presents an overview of the visual interface implementations surveyed in this section. The results from this section were synthesized into common themes and findings and presented in Table IV.

## A. MR Displays

MR is a display subclass that spans between the extrema on the reality-virtuality continuum that juxtaposes real and virtual objects within a scene [99]. Contrary to virtual environments where the user is totally immersed in and interacts with a purely synthetic world, MR displays integrate elements of both virtual and real worlds. MR displays provide a user with additional or enhanced information which can reduce operator mistakes [100], communicate processed sensor data [101], and improve human–robot collaboration [102]. MR displays are either AR, where a display of the real environment is augmented by virtual objects, or AV, where a virtual environment is augmented with real-world objects [99].

Type of Display	Summary of Findings
Augmented Reality	Task performance benefits gained from augmented coordinate system displays and depth information; calibration is important to avoid negative performance; typically used to aid in remote control or human-assisted operations. [33], [48], [103], [106]–[109], [111]
Augmented Virtuality	Can improve telepresence and situation awareness of the remote environment, if designed and calibrated appropriately. [45], [63]
Virtual Reality	Virtual reconstructions typically used for motion planning and trajectory confirmation under human- assisted or human-delegated operation; especially useful for deep sea applications when poor visibility is an issue. [9], [47], [116], [118], [120]
Depth Imagery	Generally improves task performance and end-effector placement; limitations on scenarios and tasks for which depth sensing is beneficial. [14], [125]–[127]

 TABLE IV

 Summary of the Effects of Visualization-Type on Telemanipulation Performance

1) Augmented Reality: AR interfaces display real-world imagery with synthetic object and indicator overlay cues and aids. This section focuses on monitor-based AR displays with computer-generated objects overlaid onto the imagery, as opposed to immersive environments. These object overlays, such as coordinate systems [48], [103], depth information [33], [101], or virtual handles [50], aid in manipulator positioning under remote control and human-assisted teleoperation.

Augmented coordinate system displays can improve the operator's mental model of the position and orientation of the manipulator and reduce operation errors. Chintamani *et al.* [103] found that augmenting video with the manipulator's end-effector coordinate system reduced reversal errors and total distance traveled by the end effector. Nawab *et al.* [48] generated virtual color-coded coordinate systems on the end-effector of a manipulator that mapped to similarly color-coded joysticks for controlling position and orientation, which also reduced total distance traveled and reversal errors.

Virtual handles, or 3-D widgets [104], in an AR interface alert the user to the types of allowable interactions within a remote environment to improve interaction. For example, a 3-D ring with arrows indicates potential rotation movement. Hashimoto *et al.* [50] implemented touch-based interface for a UGV manipulator with two virtual handles, including a lever to indicate manipulator movement and a ring to indicate rotation movement, which participants found intuitive to use. Chen *et al.* [105] also implemented virtual rings on an assistive ground robot interface to indicate rotation direction around identified objects.

Augmented depth information informs an operator of how far away objects are and when an object of interest is within reach of the manipulator. An early implementation of AR depth sensing was the AR through graphical overlays on stereo video remote manipulation system by Milgram *et al.* [106], [107]. Their system used virtual pointers and tape measures to calculate and display distances between user-selected points, which improved user performance [108]. Vozar and Tilbury [33] implemented a virtual crosshair on a UGV manipulator interface that changed colors to either green (if the object was within reach), or red (if the object was out of reach) to provide depth cuing. Users felt enhanced presence in the remote workspace when using the AR interface, but performance dropped compared to using a video-only interface. Depth distortions due to calibration errors can negatively affect performance when using virtual elements [109]; therefore, AR systems should draw digitally generated graphics with the same calibration parameters as the video, so users can accurately align graphic objects to real objects [110], [111]. Additionally, digitally overlaid objects will always occlude objects in the video, and interface designers should create object displays that minimize occlusion [109]. For example, instead of using solid reconstructions of remote objects (which might appear the most realistic), wireframe reconstructions reduce occlusion yet accurately represent object shape, size, and location [50], [107], [112].

2) Augmented Virtuality: AV displays use virtual environments augmented with real objects [99]. Generating AV is more computationally intensive than AR, and the quality of the virtual environment reconstruction is primarily dependent on the quality of sensor used in reconstruction [117]. Generation of virtual 3-D reconstructions of remote environments requires laser scanners [46], [63] or imagery from multiple views [118].

Moore *et al.* [46] developed a UGV interface to display a virtual ground vehicle reconstruction using stereo imagery and LiDAR data, augmented by real-world object imagery and virtual handles. Atherton and Goodrich [63] developed an AV interface for a remote manipulator to display both the virtual environment and the manipulator generated by a range imaging scanner, and augmented the virtual environment with video from a camera mounted on the end effector. Results from user experimentation included decreased mental workload and increased situation awareness; however, their AV interface increased task time, likely caused by nonoptimal calibration of the virtual elements. When designed carefully, however, virtual elements can potentially result in teleoperation performance comparable to line of sight performance [45].

#### B. Virtual Reality (VR)

VR is an extrema on the reality-virtuality continuum where the environment contains only virtual objects [99]. A VR display can improve situational awareness and partially compensate for communication delays between the vehicle and operator, especially in deep sea explorations [10], [119]. Virtual environments enable the operator to test, preview, and verify planned sequences of motion. Sensors, such as stereo cameras, range finders, or multiple camera views, are necessary to capture both depth and position information when creating a virtual environment from the robot's surroundings.

To facilitate human-assisted or human-delegated operation, virtual reconstruction environments can display a preview of predefined manipulator actions before they are physically carried out by the system. An ROV VR interface provided users with a planning tool to assist in creating and reviewing manipulator sequences on a ground vehicle [6]. Another VR interface-enabled testing and optimization of an ROV manipulator, and users noted that small manipulation behaviors during system tuning were easier to see in the VR reconstruction as opposed to on the real manipulator [120].

Virtual reconstructions are particularly helpful for visualization of ROV manipulators in underwater environments where limited viewing angle, turbidity, and poor lighting can obstruct images. A VR interface allowed operators to input the desired ROV end-effector position with a single click to generate and preview a manipulator trajectory [9]. Zhang *et al.* [116] also simulated motion command inputs in a VR environment before the ROV semiautonomously completed the movements.

VR environments are beneficial when the remote environment is unknown but can be modeled. A VR interface for the ROV developed by Marani *et al.* [114] graphically reconstructed the vehicle in a virtual environment and displayed robot and sensor information in side panels. In addition to reconstructing the ocean bottom profile, this VR environment also simulated autonomous mission planning. Hine *et al.* [10] used stereo imagery and navigational information to generate a virtual environment terrain model for deep sea sampling. In addition to virtually reconstructing unknown remote environments, VR interfaces also enable the user to interact with the system from any point of view [10], [114], [118].

Without accurate representation of the vehicle's surroundings in the virtual reconstruction, operators may find the model misleading or incomplete due to a lack of information. The interface for Nomad, a ground vehicle for desert operations [7], included a VR representation of the robot's state. Their system did not transmit detailed local terrain models, and operators choose not to view the VR reconstruction because of a lack of contextual information.

On the contrary, 3-D displays that are too information-rich can impair performance. Olmos *et al.* [121] found that when one view in a split-screen display is more information-rich (e.g., 3-D immersive environment), operators inappropriately allocate attention to that display, even when the task required attention on a different display. Similarly, Thomas and Wickens [122] observed a "cognitive tunneling" effect when participants used an immersive display compared to an egocentric display, likely caused by a failure to integrate information accurately across two different frames of reference. It has been suggested that interfaces should combine visual and auditory alerts with information-rich VR views to improve attention allocation and lessen visual loading [121], [123], [124].

### C. Stereo Vision and Depth Imaging

Stereo vision is widely used in teleoperation [128] and requires two cameras at a fixed displacement to obtain two different scene views. Many studies found that stereo 3-D displays are beneficial and indicate clear, positive performance benefits for spatial manipulations compared to monoscopic displays [125]. In particular, stereovision is beneficial for aiding in end-effector positioning in a remote workspace. Stereo video also improved deep sea exploration in unknown environments using the Virtual Environment Vehicle Interface control software developed by NASA, which enabled successful collection and sampling using a robotic ROV manipulator [10].

Depth sensors, such as time-of-flight and structured-light sensors [129], are also used to acquire and display 3-D image data in two dimensions. Active range imaging systems use light pulses and resolve distances based on the known speed of light. Day *et al.* [14] used active range sensors in a UGV bomb disposal interface to track and display distance from the end effector to objects of interest. Their interface was successfully tested by subject matter experts who found the display generated by the range sensor desirable. One limitation of active depth sensors, however, is low resolution [14], [129], [130], although time-of-flight and stereo vision data create more precise depth maps when combined [131].

Although often beneficial, stereoscopic depth displays do not always yield better performance over monoscopic displays. Drascic *et al.* [126] found that stereovision decreased task completion time only after the user gained enough operational experience when operating a mobile explosive disposal ground robot. Additionally, Draper *et al.* [127] found that stereovision was better only under difficult task conditions when assessing task performance. Additionally, limited research exists comparing the usage of imagery from stereo vision to active time-of-flight range sensors for performing manipulation tasks.

## V. DISCUSSION AND CONCLUSION

The purpose of this article was to collate, summarize, and discuss literature focused on HMI for performing telemanipulation tasks using SUS and identify areas where further research is needed. This article focused only on the telemanipulation task for systems with small work environments, and did not include navigation tasks or tasks with time delays. This article included interfacing technologies and implementations from over 70 different teleoperated SUS manipulator systems. Tables I and III included summaries of physical and perceptual technologies commonly used for telemanipulation, while Tables II and IV presented a summary of findings from the surveyed literature. In addition to presenting the interface technologies and input devices for each system, robot level of autonomy was also included in this analysis as it affects human-robot interaction [22]. In all cases, an operator assumes the role of being primarily responsible for performing or monitoring telemanipulation tasks in a remote environment, which provided a basis for comparison.

As illustrated in Tables I and III, a majority of telemanipulating SUS in the literature operate under remote control. The use of remote control manipulators was likely due to the unstructured nature of operational environments for SUS, for example, in search and rescue missions. When systems cannot know the remote environment a priori, manual or computer-aided teleoperation becomes a default operation mode. A rich history of literature exists on stationary manual teleoperation HMI, and much of that work has been applied to SUS telemanipulation, for example, the use of scaled down master-slave controllers and force feedback. Different types of advanced visualization aid in direct telemanipulation, with a majority of systems relying on some form of real-world imagery through a live video feed. Additionally, studies have applied previous telemanipulation findings to recent technologies, including end-effector control implementations on workstations and tablets, and force feedback through portable, small hand controllers, although more work is needed to conduct focused, field-based HMI research using these technologies.

Physical control input devices are an important link between the remote system and human operator, and form factor of the device facilitates the type and autonomy of interaction with the system. Nearly all master-slave and joystick controllers (and other variations of physical controllers) only allow for direct remote control or human-assisted operation, which require near constant input from the operator. Control should generally be at the end effector when operating under manual teleoperation, instead of each individual joint. While continuous teleoperation using these systems can be fatiguing, control enhancements such as haptic feedback or assistive virtual fixtures improve overall performance. Physical devices with computing power, such as laptops or tablets, afford goal-oriented input and directives from the operator when systems have advanced autonomous capabilities. Virtual fixtures allow humans to safely cope with more complex and unstructured tasks and remain within the control loop, but adaptive fixtures require additional research to develop constraints in near real time, as this article was in its early stages [96]. Additionally, the past 5-10 years have seen a trend of increased mobile tablet use for controlling teleoperated systems, which will likely continue to increase as computing power and connectivity evolve; however, further evaluation of touch-based gesturing and interaction is needed to develop effective manipulator direct control through mobile devices.

Visual enhancements using processed sensor data offer great potential for improving telemanipulation performance of SUS. Depth sensing through stereoscopic or time-of-flight sensors can enhance overall task performance and end-effector placement. Visual cues using AR (such as coordinate system displays or virtual handles) offer task performance benefits during direct control operations with minimal computational effort; however, the virtual objects should be calibrated appropriately. Complete VR environments, typically used for motion planning and trajectory confirmation, are effective for systems operating at levels of autonomy higher than remote control, especially when poor visibility of the remote environment is prohibitive. It is recommended that additional research compare AV to both VR and AR, as researchers suggest AV offers the benefits of both virtual reconstructions and real-world imagery and makes relationships in the environment perceptually evident to the user [63]. Overall,

the quality and type of visual display greatly affect operator performance, and the appropriate type of visual display depends on SUS capabilities, manipulation task, and level of autonomy.

Operators can take on a more supervisory role while in the loop, but a majority of tasks still require some form of direct teleoperation control. Sheridan [132] notes that "all robots for the foreseeable future will be controlled by humans, either as teleoperators steered by continuous manual movement or as telerobots intermittently monitored and reprogrammed by human supervisors." A large proportion of focused telemanipulation SUS studies used remote control or human-assisted systems, but it is evident that the autonomous capabilities of manipulating SUS are advancing from remote control toward human-supervised systems. Experimentation is needed to verify that manipulating SUS have the necessary technical maturity and fit the requirements of desired missions with higher levels of autonomy [133]. Additionally, instead of replacing a human with a fully autonomous robot, focus should shift toward designing HMI to facilitate human-supervised interaction. High-fidelity experiments are recommended to understand the appropriate interfacing, controls, and level of operator interaction as telemanipulation missions shift toward higher levels of autonomy.

#### REFERENCES

- J. Peschel, "Towards physical object manipulation by small unmanned aerial systems," in *Proc. IEEE Int. Symp. Saf., Secur., Rescue Robot.*, College Station, TX, USA, 2012, pp. 1–6.
- [2] Department of Defense UAS Task Force, Airspace Integration Integrated Product Team, "Department of Defense unmanned aircraft system airspace integration plan, Version 2.0," United States Department of Defense, Washington, DC, USA, Tech. Rep. 1-7ABA52E, 2011.
- [3] Q. Lindsey, D. Mellinger, and P. Kumar, "Construction of cubic structures with quadrotor teams," in *Proc. Robot.: Sci. Syst. VII*, Los Angeles, CA, USA, 2011, pp. 177–184.
- [4] J. Werfel, Y. Bar-Yam, D. Rus, and R. Nagpal, "Distributed construction by mobile robots with enhanced building blocks," in *Proc. IEEE Int. Conf. Robot. Autom.* Orlando, FL, USA, 2006, pp. 2787–2794.
- [5] Q. Lindsey, D. Mellinger, and V. Kumar, "Construction with quadrotor teams," *Auton. Robots*, vol. 33, no. 3, pp. 323–336, 2012.
- [6] L. A. Nguyen *et al.*, "Virtual reality interfaces for visualization and control of remote vehicles," *Auton. Robots*, vol. 11, no. 1, pp. 59–68, 2001.
- [7] D. Wettergreen *et al.*, "Operating Nomad during the Atacama Desert Trek," in *Field Service Robotics*, A. Zelinsky, Ed., London, U.K.: Springer, 1998, pp. 82–89.
- [8] A. Parness *et al.*, "Gravity-independent rock-climbing robot and a sample acquisition tool with microspine grippers," *J. Field Robot.*, vol. 30, no. 6, pp. 897–915, 2013.
- [9] H. Shim, B. H. Jun, P. M. Lee, H. Baek, and J. Lee, "Workspace control system of underwater tele-operated manipulators on an ROV," *Ocean. Eng.*, vol. 37, no. 11/12, pp. 1036–1047, 2010.
- [10] B. Hine *et al.*, "The application of telepresence and virtual reality to subsea exploration," in *Proc. 2nd Workshop Mobile Robots Subsea Environ.*, Monterey, CA, USA, 1994. [Online]. Available: https://ri.cmu.edu/ pub\_files/pub1/iii\_b\_p\_hine\_1994\_1/iii\_b\_p\_hine\_1994\_1.pdf
- [11] C. J. Cannell and D. J. Stilwell, "A comparison of two approaches for adaptive sampling of environmental processes using autonomous underwater vehicles," in *Proc IEEE*. OCEANS MTS/IEEE, 2005, pp. 1514–1521.
- [12] M. Dunbabin and L. Marques, "Robots for environmental monitoring: Significant advancements and applications," *IEEE Robot. Autom. Mag.*, vol. 19, no. 1, pp. 24–39, Mar. 2012.
- [13] M. Jadaliha and J. Choi, "Environmental monitoring using autonomous aquatic robots: Sampling algorithms and experiments," *IEEE Trans. Control Syst. Technol.*, vol. 21, no. 3, pp. 899–905, May 2013.

- [14] B. Day, C. Bethel, R. Murphy, and J. Burke, "A depth sensing display for bomb disposal robots," in *Proc. IEEE Int. Workshop Saf., Secur. Rescue Robot.*, Sendai, Japan, 2008, pp. 146–151.
- [15] M. Mast *et al.*, "Semi-autonomous domestic service robots: Evaluation of a user interface for remote manipulation and navigation with focus on effects of stereoscopic display," *Int. J. Social Robot.*, vol. 7, no. 2, pp. 183–202, 2015.
- [16] D. Katz, A. Venkatraman, M. Kazemi, J. A. Bagnell, and A. Stentz, "Perceiving, learning, and exploiting object affordances for autonomous pile manipulation," *Auton. Robots*, vol. 37, no. 4, pp. 369–382, 2014.
- [17] T. M. Howard, M. Pivtoraiko, R. A. Knepper, and A. Kelly, "Modelpredictive motion planning: Several key developments for autonomous mobile robots," *IEEE Robot. Autom. Mag.*, vol. 21, no. 1, pp. 64–73, Mar. 2014.
- [18] M. Naveau, M. Kudruss, O. Stasse, C. Kirches, K. Mombaur, and P. Souères, "A reactive walking pattern generator based on nonlinear model predictive control," *IEEE Robot. Autom. Lett.*, vol. 2, no. 1, pp. 10–17, Jan. 2017.
- [19] C. C. Kemp, A. Edsinger, and E. Torres-Jara, "Challenges for robot manipulation in human environments [Grand challenges of robotics]," *IEEE Robot. Autom. Mag.*, vol. 14, no. 1, pp. 20–29, Mar. 2007.
- [20] C. M. Dellin, K. Strabala, G. C. Haynes, D. Stager, and S. S. Srinivasa, "Guided manipulation planning at the DARPA robotics challenge trials," in *Experimental Robotics*, M. A. Hsieh, O. Khatib, and V. Kumar, Eds., Marrakech, Morocco: Springer, 2016, pp. 149–163.
- [21] M. A. Goodrich and A. Schultz, "Human-robot interaction: A survey," *Found. Trends Human-Comput. Interact.*, vol. 1, no. 3, pp. 203–275, 2007.
- [22] J. M. Beer, A. D. Fisk, and W. A. Rogers, "Toward a framework for levels of robot autonomy in human-robot interaction," *J. Human-Robot Interact.*, vol. 3, no. 2, pp. 74–99, 2014.
- [23] R. Parasuraman, T. B. Sheridan, and C. D. Wickens, "A model for types and levels of human interaction with automation," *IEEE Trans. Syst.*, *Man, Cybern. A, Syst. Humans*, vol. 30, no. 3, pp. 286–297, May 2000.
- [24] G. Niemeyer, C. Preusche, and G. Hirzinger, "Telerobotics," in *Springer Handbook of Robotics*. Berlin, Germany: Springer, 2008, pp. 741–757.
- [25] H.-M. Huang, K. Pavek, J. Albus, and E. Messina, "Autonomy levels for unmanned systems (ALFUS) framework: An update," in *Proc. Unmanned Ground Vehicle Technol. VII*, 2005, vol. 5804, pp. 439–449.
- [26] T. B. Sheridan, *Telerobotics, Automation, and Human Supervisory Con*trol. Cambridge, MA, USA: MIT Press, 1992.
- [27] L. Slutski, "Control methods in the systems 'operator-remote manipulator'," in *Remote Manipulation Systems: Quality Evaluation and Improvement*, (Microprocessor-Based and Intelligent Systems Engineering), S. Tzafestas, Ed., Norwell, MA, USA: Kluwer, 1998, vol. 17, pp. 59–117.
- [28] M. Hildebrandt, L. Christensen, J. Kerdels, J. Albiez, and F. Kirchner, "Realtime motion compensation for ROV-based tele-operated underwater manipulators," in *Proc. OCEANS-EUROPE*, Bremen, Germany, 2009, pp. 1–6.
- [29] B.-H. Jun, P. M. Lee, and J. Lee, "Manipulability analysis of underwater robotic arms on ROV and application to task-oriented joint configuration," in *Proc. MTS/IEEE TECHNO-OCEAN*, Kobe, Japan, Nov. 2004, vol. 3, pp. 1548–1553.
- [30] S. McMillan, D. E. Orin, and R. B. McGhee, "Efficient dynamic simulation of an underwater vehicle with a robotic manipulator," *IEEE Trans. Syst., Man, Cybern.*, vol. 25, no. 8, pp. 1194–1206, Aug. 1995.
- [31] Seamor Marine, "Seamor ROV Specifications," Accessed: Oct. 15, 2017. [Online]. Available: http:///seamor.com/products/
- [32] R. R. Murphy et al., "Use of remotely operated marine vehicles at Minamisanriku and Rikuzentakata Japan for disaster recovery," in Proc. IEEE Int. Symp. Saf., Secur. Rescue Robot., Kyoto, Japan, Nov. 2011, pp. 19–25.
- [33] S. Vozar and D. M. Tilbury, "Augmented reality user interface for mobile robots with manipulator arms: Development, testing, and qualitative analysis," in *Proc. ASME Int. Des. Eng. Tech. Conf. Comput. Inf. Eng. Conf.*, Chicago, IL, USA, 2012, pp. 1341–1350.
- [34] G. A. Nixon, J. G. Wildenbeest, and D. A. Abbink, "Effective humanmachine interfaces for aerial telemanipulation," *J. Delft Haptics Lab*, vol. 1, no. 1, pp. 1–10, 2015.
- [35] R. Clavel, "Device for the movement and positioning of an element in space," U.S. Patent 4 976 582, 1989.
- [36] S. Martin and N. Hillier, "Characterization of the Novint Falcon haptic device for application as a robot manipulator," in *Proc. Australasian Conf. Robot. Autom.*, Sydney, Australia, 2009, pp. 1–9.

- [37] F. Schill, R. Mahony, P. Corke, and L. Cole, "Virtual force feedback teleoperation of the InsectBot using optic flow," in *Proc. Australasian Conf. Robot. Autom.*, Canberra, Australia, 2008, pp. 1–9.
- [38] D. Song, K. Goldberg, and N. Y. Chong, "Networked telerobots," in *Springer Handbook of Robotics*. Berlin, Germany: Springer, 2008, pp. 759–771.
- [39] I. Belousov, S. Chebukov, and V. Sazonov, "Web-based teleoperation of the robot interacting with fast moving objects," in *Proc. IEEE Int. Conf. Robot. Autom.*, Barcelona, Spain, 2005, pp. 673–678.
- [40] A. K. Mishra and O. Meruvia-Pastor, "Robot arm manipulation using depth-sensing cameras and inverse kinematics," in *Proc. Oceans - St. John's.*, St. John's, NL, Canada, 2014, pp. 1–6.
- [41] T. L. Chen et al., "Robots for humanity: A case study in assistive mobile manipulation," *IEEE Robot. Autom. Mag., Special Issue Assistive Robot.*, vol. 20, no. 1, 2013. [Online]. Available: http://www.willowgarage.com/ sites/default/files/Chen\_RFH\_ram\_2013.pdf
- [42] A. Speers, P. M. Forooshani, M. Dicke, and M. Jenkin, "Lightweight tablet devices for command and control of ROS-enabled robots," in *Proc. 16th Int. Conf. Adv. Robot.*, Montevideo, Uruguay, Nov. 2013, pp. 1–6.
- [43] Z. Materna *et al.*, "Teleoperating assistive robots: A novel user interface relying on semi-autonomy and 3D environment mapping," *J. Robot. Mechatronics*, vol. 29, no. 2, pp. 381–394, 2017.
- [44] S. Vozar, "A framework for improving the speed and performance of teleoperated mobile manipulators," Ph.D. dissertation, Dept. Mech. Eng., Univ. Michigan, Ann Arbor, MI, USA, 2013.
- [45] C. S. Tzafestas, N. Palaiologou, and M. Alifragis, "Experimental evaluation and pilot assessment study of a virtual and remote laboratory on robotic manipulation," in *Proc. IEEE Int. Symp. Ind. Electron.*, Dubrovnik, Croatia, Jun. 2005, vol. 4, pp. 1677–1684.
- [46] J. Moore *et al.*, "Nested marsupial robotic system for search and sampling in increasingly constrained environments," in *Proc. IEEE Int. Conf. Sys.*, *Man, Cybern.*, Budapest, Hungary, Oct. 2016, pp. 279–286.
- [47] H. G. Nguyen and J. P. Bott, "Robotics for law enforcement: Applications beyond explosive ordnance disposal," in *Proc. Enabling Technol. Law Enforcement Secur.*, 2001, vol. 4232, pp. 433–455.
- [48] A. Nawab, K. Chintamani, D. Ellis, G. Auner, and A. Pandya, "Joystick mapped augmented reality cues for end-effector controlled tele-operated robots," in *Proc. IEEE Virtual Reality Conf.*, Costa Mesa, CA, USA, 2007, pp. 263–266.
- [49] Y. P. Toh, S. Huang, J. Lin, M. Bajzek, G. Zeglin, and N. S. Pollard, "Dexterous telemanipulation with a multi-touch interface," in *Proc. Proc. 12th IEEE-RAS Int. Conf. Humanoid Robots*, Osaka, Japan, 2012, pp. 270–277.
- [50] S. Hashimoto, A. Ishida, M. Inami, and T. Igarashi, "Touchme: An augmented reality based remote robot manipulation," in *Proc. 21st Int. Conf. Artif. Reality Telexistence*, Osaka, Japan, 2011, vol. 2, pp. 529–537.
- [51] D. A. Lopez, J. A. Frank, and V. Kapila, "Comparing interface elements on a tablet for intuitive teleoperation of a mobile manipulator," in *Proc. ASME 12th Biennial Conf. Eng. Syst. Des. Anal.*, Copenhagen, Denmark, 2014, pp. 1–10.
- [52] C. Parga, X. Li, and W. Yu, "Tele-manipulation of robot arm with smartphone," in *Proc. Proc. 6th Int. Symp. Resilient Control Syst.*, San Francisco, CA, USA, 2013, pp. 60–65.
- [53] J. V. Draper, "Teleoperator hand controllers: A contextual human factors assessment," in *Environmental Restoration and Waste Management Program.* Oak Ridge, TN, USA: Oak Ridge National Laboratory, 1994.
- [54] T. B. Sheridan, "Supervisory control of remote manipulators, vehicles, and dynamic processes: Experiments in command and display aiding," Eng. Psychol. Group Office Naval Res., Arlington, Virginia, Tech. Rep. DTIC ADA129678, 1983.
- [55] T. B. Sheridan and W. L. Verplank, "Human and computer control of undersea teleoperators," Cambridge Man-Machine Syst. Lab., Massachusetts Inst. Technol., Cambridge, MA, USA, Tech. Rep. ADA057655, 1978.
- [56] O. Ben-Porat, M. Shoham, and J. Meyer, "Control design and task performance in endoscopic teleoperation," *Presence*, vol. 9, no. 3, pp. 256–267, Jun. 2000.
- [57] F. Lai, R. Howe, and P. Millman, "Frame mapping and dexterity for surgical task performance in robotic endoscopic surgery," in *Proc. ASME Dyn. Sys. Control Div.*, Nashville, TN, USA, 1999.
- [58] M. R. Endsley, "Design and evaluation for situation awareness enhancement," *Proc. Human Factors Soc. Annu. Meeting*, vol. 32, no. 2, pp. 97–101, 1988.

- [59] E. You and K. Hauser, "Assisted teleoperation strategies for aggressively controlling a robot arm with 2D input," in *Proc. Robot.: Sci. Syst.*, Los Angeles, CA, USA, 2011, pp. 354–361.
- [60] A. Singh, S. H. Seo, Y. Hashish, M. Nakane, J. E. Young, and A. Bunt, "An interface for remote robotic manipulator control that reduces task load and fatigue," in *Proc. IEEE Conf. Robot Human Interactive Commun.*, Gyeongju, South Korea, Aug. 2013, pp. 738–743.
- [61] P. Hebert *et al.*, "Supervised remote robot with guided autonomy and teleoperation (SURROGATE): A framework for whole-body manipulation," in *Proc. IEEE Int. Conf. Robot. Autom.*, Seattle, WA, USA, May 2015, pp. 5509–5516.
- [62] G. Lunghi, R. M. Prades, and M. D. Castro, "An advanced, adaptive and multimodal graphical user interface for human-robot teleoperation in radioactive scenarios," in *Proc. 13th Int. Conf. Inform. Control, Autom. Robot.*, 2016, pp. 224–231.
- [63] J. A. Atherton and M. A. Goodrich, "Supporting remote manipulation with an ecological augmented virtuality interface," in *Proc. Symp. New Frontiers Human-Robot Interact./Artif. Intell. Simul. Behaviour*, 2009. [Online]. Available: http://citeseerx.ist.psu.edu/viewdoc/ download?doi=10.1.1.159.8634&rep=rep1&type=pdf
- [64] M. Bengel, K. Pfeiffer, B. Graf, A. Bubeck, and A. Verl, "Mobile robots for offshore inspection and manipulation," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, St. Louis, MO, USA, 2009, pp. 3317–3322.
- [65] A. H. Herdocia, A. Shademan, and M. Jägersand, "Unimodal asymmetric interface for teleoperation of mobile manipulators: A user study," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Vilamoura, Portugal, 2012, pp. 5214–5219.
- [66] A. B. Kuang, S. Payandeh, Z. Bin, F. Henigman, and C. L. MacKenzie, "Assembling virtual fixtures for guidance in training environments," in *Proc. 12th Int. Symp. Haptic Interfaces Virtual Environ. Teleoperator Syst.*, Chicago, IL, USA, 2004, pp. 367–374.
- [67] P. M. Lee *et al.*, "System design and development of a deep-sea unmanned underwater vehicle HEMIRE for oceanographic research," in *Proc. 6th Int. Offshore Polar Eng. Conf.*, Los Angeles, CA, USA, 2006, pp. 205–212.
- [68] S. E. Navarro, F. Heger, F. Putze, T. Beyl, T. Schultz, and B. Hein, "Telemanipulation with force-based display of proximity fields," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Hamburg, Germany, 2015, pp. 4568–4574.
- [69] S. Payandeh and Z. Stanisic, "On application of virtual fixtures as an aid for telemanipulation and training," in *Proc. 10th Symp. Haptic Interfaces Virtual Environ. Teleoperator Syst.*, Orlando, FL, USA, 2002, pp. 18–23.
- [70] F. Saito and T. Suehiro, "Toward telemanipulation via 2-D interfaceconcept and first result of "Titi"," in *Proc. IEEE 28th Annu. Conf. Ind. Electron. Soc.*, Sevilla, Spain, 2002, vol. 3, pp. 2243–2248.
- [71] S. Vozar and D. M. Tilbury, "Improving UGV teleoperation performance using novel visualization techniques and manual interfaces," *Proc SPIE*, vol. 8387, May 25, 2012, Art. no. 838 716.
- [72] T. Xia, S. Léonard, A. Deguet, L. Whitcomb, and P. Kazanzides, "Augmented reality environment with virtual fixtures for robotic telemanipulation in space," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Vilamoura, Portugal, 2012, pp. 5059–5064.
- [73] M. C. Yip, M. Tavakoli, and R. D. Howe, "Performance analysis of a haptic telemanipulation task under time delay," *Adv. Robot.*, vol. 25, no. 5, pp. 651–673, 2011.
- [74] R. Jagacinski, D. Repperger, S. Ward, and M. Moran, "A test of Fitts' law with moving targets," *Human Factors*, vol. 22, pp. 225–233, 1980.
- [75] R. Jagacinski, D. Repperger, M. Moran, S. Ward, and B. Glass, "Fitts" law and the microstructure of rapid discrete movements," *J. Exp. Psychol. Human Perception Perform.*, vol. 6, pp. 309–320, 1980.
- [76] B. Kantowitz and G. Elvers, "Fitts' law with an isometric controller: Effects of order of control and control-display gain." J. Motor Behav., vol. 20, no. 1, pp. 53–66, 1988.
- [77] K. Won, F. Tendick, S. Ellis, and L. Stark, "A comparison of position and rate control for telemanipulations with consideration of manipulator system dynamics," *IEEE J. Robot. Autom.*, vol. 3, no. 5, pp. 426–436, Oct. 1987.
- [78] G. S. Gupta, S. C. Mukhopadhyay, C. H. Messom, and S. N. Demidenko, "Master-slave control of a teleoperated anthropomorphic robotic arm with gripping force sensing," *IEEE Trans. Instrum. Meas.*, vol. 55, no. 6, pp. 2136–2145, Dec. 2006.

- [79] K. S. Hale and K. M. Stanney, "Deriving haptic design guidelines from human physiological, psychophysical, and neurological foundations," *IEEE Comput. Graph. Appl.*, vol. 24, no. 2, pp. 33–39, Mar./Apr. 2004.
- [80] V. Nitsch and B. Färber, "A meta-analysis of the effects of haptic interfaces on task performance with teleoperation systems," *IEEE Trans. Haptics*, vol. 6, no. 4, pp. 387–398, Oct./Dec. 2013.
- [81] R. P. Khurshid, N. T. Fitter, E. A. Fedalei, and K. J. Kuchenbecker, "Effects of grip-force, contact, and acceleration feedback on a teleoperated pick-and-place task," *IEEE Trans. Haptics*, vol. 10, no. 1, pp. 40–53, Jan. 2017.
- [82] L. B. Rosenberg, "The use of virtual fixtures as perceptual overlays to enhance operator performance in remote environments," Stanford Univ. Center Des. Res., Stanford, CA, USA, Tech. Rep. ADA292450, 1992.
- [83] L. Ming and A. M. Okamura, "Recognition of operator motions for real-time assistance using virtual fixtures," in *Proc. 11th Symp. Haptic Interfaces Virtual Environ. Teleoperator Syst.*, Los Angeles, CA, USA, 2003, pp. 125–131.
- [84] J. J. Abbott, P. Marayong, and A. M. Okamura, "Haptic virtual fixtures for robot-assisted manipulation," in *Robotics Research*, S. Thrun, R. Brooks, and H. Durrant-Whyte, Eds., Berlin, Germany: Springer-Verlag, 2007, pp. 49–64.
- [85] W. Kim, F. Tendick, and L. Stark, "Visual enhancements in pickand-place tasks: Human operators controlling a simulated cylindrical manipulator," *IEEE J. Robot. Autom.*, vol. 3, no. 5, pp. 418–425, Oct. 1987.
- [86] D. A. Lawrence, "Stability and transparency in bilateral teleoperation," *IEEE Trans. Robot. Autom.*, vol. 9, no. 5, pp. 624–637, Oct. 1993.
- [87] S. E. Salcudean, M. Zhu, W.-H. Zhu, and K. Hashtrudi-Zaad, "Transparent bilateral teleoperation under position and rate control," *Int. J. Robot. Res.*, vol. 19, no. 12, pp. 1185–1202, Dec. 2000.
- [88] T. W. Kim, G. Marani, and J. Yuh, "Underwater vehicle manipulators," in *Springer Handbook of Ocean Engineering*, M. R. Dhanak and N. I. Xiros, Eds., Berlin, Germany: Springer-Verlag, 2016, pp. 407–422.
- [89] L. J. Love and W. J. Book, "Force reflecting teleoperation with adaptive impedance control," *IEEE Trans. Syst., Man, Cybern., B. Cybern.*, vol. 34, no. 1, pp. 159–165, Feb. 2004.
- [90] P. Fischer, R. Daniel, and K. Siva, "Specification and design of input devices for teleoperation," in *Proc. IEEE Int. Conf. Robot. Autom.*, Cincinnati, OH, USA, 1990, pp. 540–545.
- [91] T. N. Judkins, D. Oleynikov, K. Narazaki, and N. Stergiou, "Robotic surgery and training: Electromyographic correlates of robotic laparoscopic training," *Surg. Endosc.*, vol. 20, no. 5, pp. 824–829, 2006.
- [92] A. M. Okamura, "Methods for haptic feedback in teleoperated robotassisted surgery," *Ind. Robot.*, vol. 31, no. 6, pp. 499–508, 2004.
- [93] B. Weber and C. Eichberger, "The benefits of haptic feedback in telesurgery and other teleoperation systems: A meta-analysis," in *Proc. Int. Conf. Universal Access Human-Comput. Interact.*, Los Angeles, CA, USA, 2015, pp. 394–405.
- [94] L. B. Rosenberg, "Virtual fixtures as tools to enhance operator performance in telepresence environments," in *Proc. Telemanipulator Technol. Space Telerobotics*, 1993, vol. 2057, pp. 10–22.
- [95] A. Bettini, S. Lang, A. Okamura, and G. Hager, "Vision assisted control for manipulation using virtual fixtures: Experiments at macro and micro scales," in *Proc. IEEE Int. Conf. Robot. Autom.*, Washington, DC, USA, 2002, vol. 4, pp. 3354–3361.
- [96] S. A. Bowyer, B. L. Davies, and F. R. y Baena, "Active constraints/virtual fixtures: A survey," *IEEE Trans. Robot.*, vol. 30, no. 1, pp. 138–157, Feb. 2014.
- [97] F. Rydén, H. J. Chizeck, S. N. Kosari, H. King, and B. Hannaford, "Using Kinect and a haptic interface for implementation of real-time virtual fixtures," in *Proc. 2nd Workshop RGB-D: Adv. Reasoning With Depth Cameras (in Conjunction With RSS)*, Los Angeles, CA, USA, 2011. [Online]. Available: https://pdfs.semanticscholar.org/1707/ 049190395eb0c655fa2c0cc8367c2e8417c1.pdf
- [98] D. Aarno, S. Ekvall, and D. Kragic, "Adaptive virtual fixtures for machine-assisted teleoperation tasks," in *Proc. IEEE Int. Conf. Robot. Autom.*, Barcelona, Spain, 2005, pp. 1139–1144.
- [99] P. Milgram and F. Kishino, "A taxonomy of mixed reality visual displays," *IEICE Trans. Inf. Syst.*, vol. E77-D, no. 12, pp. 1321–1329, 1994.
- [100] J. Carlson and R. R. Murphy, "How UGVs physically fail in the field," *Trans. Robot.*, vol. 21, no. 3, pp. 423–437, 2005.
- [101] S. Thomas, "Design of a teleoperated rock sampling system," Master's thesis, Dept. Mech. Eng., Virginia Polytechnic Inst. State Univ., Blacksburg, VA, USA, 2011.

- [102] S. A. Green, M. Billinghurst, X. Chen, and J. G. Chase, "Human-robot collaboration: A literature review and augmented reality approach in design," *Int. J. Adv. Robot. Syst.*, vol. 5, no. 1, 2008, Art. no. 1.
- [103] K. Chintamani, A. Cao, R. D. Ellis, and A. K. Pandya, "Improved telemanipulator navigation during display-control misalignments using augmented reality cues," *IEEE Trans. Syst., Man, Cybern., A Syst. Humans*, vol. 40, no. 1, pp. 29–39, Jan. 2010.
- [104] B. D. Conner, S. S. Snibbe, K. P. Herndon, D. C. Robbins, R. C. Zeleznik, and A. Van Dam, "Three-dimensional widgets," in *Proc. Symp. Interactive 3D Graph.*, San Diego, CA, USA, 1992, pp. 183–188.
- [105] H. Chen, Y. Kakiuchi, M. Saito, K. Okada, and M. Inaba, "View-based multi-touch gesture interface for furniture manipulation robots," in *Proc. Adv. Robot. Social Impacts*, 2011, pp. 39–42.
- [106] P. Milgram, S. Zhai, D. Drascic, and J. Grodski, "Applications of augmented reality for human-robot communication," in *Proc. IEEE/RSJ Int. Conf. Intell. Robots Syst.*, Tokyo, Japan, 1993, vol. 3, pp. 1467–1472.
- [107] P. Milgram, S. Yin, and J. J. Grodski, "An augmented reality based teleoperation interface for unstructured environments," in *Proc. Amer. Nucl. Soc. 7th Topical Meeting Robot. Remote Syst.*, 1997, pp. 966–973.
- [108] K. Ruffo and P. Milgram, "Effect of stereographic + stereovideo 'tether' enhancement for a peg-in-hole task," in *Proc. IEEE Int. Conf. Syst., Man Cybern.*, 1992, pp. 1425–1430.
- [109] D. Drascic and P. Milgram, "Perceptual issues in augmented reality," *Proc. SPIE*, vol. 2653, pp. 123–135 Apr. 1996.
- [110] D. Drascic and P. Milgram, "Positioning accuracy of a virtual stereographic pointer in a real stereoscopic video world," *Proc. SPIE*, vol. 1457, pp. 302–314 Aug. 1991.
- [111] P. Milgram, A. Rastogi, and J. J. Grodski, "Telerobotic control using augmented reality," in *Proc. 4th IEEE Int. Workshop Robot Human Commun.*, Tokyo, Japan, 1995, pp. 21–29.
- [112] A. Rastogi, "Design of an interface for teleoperation in unstructured environments using augmented reality displays," Ph.D. dissertation, Dept. Ind. Eng., Univ. Toronto, Toronto, ON, Canada, 1997.
- [113] A. Afaq et al., "Development of FPGA-based system for control of an unmanned ground vehicle with 5-DOF robotic arm," in Proc. 15th Int. Conf. Control, Autom. Syst., Bremen, Germany, Oct. 2015, pp. 724–729.
- [114] G. Marani, S. K. Choi, and J. Yuh, "Underwater autonomous manipulation for intervention missions AUVs," *Ocean Eng.*, vol. 36, no. 1, pp. 15–23, 2009.
- [115] B. H. Wilcox *et al.*, "Athlete: A cargo handling and manipulation robot for the moon," *J. Field Robot.*, vol. 24, no. 5, pp. 421–434, 2007.
- [116] Z. Ping, E. Shimizu, and M. Ito, "A teleoperating system for underwater manipulator to practice a time limit task," in *Proc. IEEE Int. Symp. Intell. Control.* Houston, TX, USA, 2003, pp. 467–471.
- [117] C. Nielsen, "Using augmented virtuality to improve human-robot interactions." Ph.D. dissertation, Dept. Compt. Sci., Brigham Young Univ., Provo, UT, USA, 2006.
- [118] R. Marin, P. J. Sanz, P. Nebot, and R. Wirz, "A multimodal interface to control a robot arm via the web: A case study on remote programming," *IEEE Trans. Ind. Electron.*, vol. 52, no. 6, pp. 1506–1520, Dec. 2005.

- [119] A. Monferrer and D. Bonyuet, "Cooperative robot teleoperation through virtual reality interfaces," in *Proc. 6th IEEE Int. Conf. Inf. Visualization*, Boston, MA, USA, 2002, pp. 243–248.
- [120] D. M. Lane *et al.*, "AMADEUS: Advanced manipulation for deep underwater sampling," *IEEE Robot. Autom. Mag.*, vol. 4, no. 4, pp. 34–45, Dec. 1997.
- [121] O. Olmos, C. D. Wickens, and A. Chudy, "Tactical displays for combat awareness: An examination of dimensionality and frame of reference concepts and the application of cognitive engineering," *Int. J. Aviation Psychol.*, vol. 10, no. 3, pp. 247–271, 2000.
- [122] L. C. Thomas and C. D. Wickens, "Visual displays and cognitive tunneling: Frames of reference effects on spatial judgments and change detection," in *Proc. Human Factors Ergonom. Soc. Annu. Meeting*, Los Angeles, CA, USA, 2001, vol. 45, pp. 336–340.
- [123] J. Casper and R. R. Murphy, "Human-robot interactions during the robotassisted urban search and rescue response at the World Trade Center," *IEEE Trans. Syst., Man, Cybern., Cybern.*, vol. 33, no. 3, pp. 367–385, Jun. 2003.
- [124] J. Y. C. Chen, E. Hass, and M. J. Barnes, "Human performance issues and user interface design for teleoperated robots," *IEEE Trans. Syst., Man, Cybern., C Appl. Rev.*, vol. 37, no. 6, pp. 1231–1245, Nov. 2007.
- [125] J. P. McIntire, P. R. Havig, and E. E. Geiselman, "Stereoscopic 3D displays and human performance: A comprehensive review," *Displays*, vol. 35, no. 1, pp. 18–26, 2014.
- [126] D. Drascic, P. Milgram, and J. Grodski, "Learning effects in telemanipulation with monoscopic versus stereoscopic remote viewing," in *Proc. IEEE Int. Conf. Syst., Man Cybern.*, Cambridge, MA, USA, 1989, vol. 3, pp. 1244–1249.
- [127] J. V. Draper, S. Handel, C. C. Hood, and C. T. Kring, "Three experiments with stereoscopic television: When it works and why," in *Proc. IEEE Int. Conf. Syst., Man, Cybern.*, Charlottesville, VA, USA, 1991, vol. 2, pp. 1047–1052.
- [128] M. Ferre, R. Aracil, and M. A. Sanchez-Uran, "Stereoscopic human interfaces," *IEEE Robot. Autom. Mag.*, vol. 15, no. 4, pp. 50–57, Dec. 2008.
- [129] A. M. Pinto, P. Costa, A. P. Moreira, L. F. Rocha, G. Veiga, and E. Moreira, "Evaluation of depth sensors for robotic applications," in *Proc. IEEE Int. Conf. Auton. Robot Syst. Competitions*, Torres Vedras, Portugal, 2015, pp. 139–143.
- [130] S. Foix, G. Alenya, and C. Torras, "Lock-in time-of-flight (ToF) cameras: A survey," *IEEE Sensors J.*, vol. 11, no. 9, pp. 1917–1926, Sep. 2011.
- [131] Z. Jiejie, W. Liang, Y. Ruigang, and J. Davis, "Fusion of time-of-flight depth and stereo for high accuracy depth maps," in *Proc. IEEE Conf. Comput. Vision Pattern Recognit.*, Anchorage, AK, USA, 2008, pp. 1–8.
- [132] T. B. Sheridan, "Human-robot interaction: Status and challenges," *Human Factors*, vol. 58, no. 4, pp. 525–532, 2016.
- [133] B. A. Duncan and R. R. Murphy, "Autonomous capabilities for small unmanned aerial systems conducting radiological response: Findings from a high-fidelity discovery experiment," *J. Field Robot.*, vol. 31, no. 4, pp. 522–536, 2014.